1 Trends and regional variations in prescriptions dispensed to stimulate uterine

2 contractions at the end of pregnancy in Belgium: a community-based study

3 from 2003 to 2018

4 Running title: Peripartum uterotonics prescribed in Belgium

- 5 Lionel Larcin ¹ Güngör Karakaya ^{2,3} Xavier Rygaert ² Philippe Van Wilder ⁴ Clotilde Lamy ⁵ Bart
- 6 Demyttenaere ^{2,6} Christine Damase-Michel ⁷ Fati Kirakoya-Samadoulougou ¹
- 7 1 Centre de recherche Epidémiologie, Biostatistique et Recherche Clinique, Ecole de santé publique,
- 8 Université libre de Bruxelles (U.L.B.), Bruxelles, Belgium
- 9 2 Agence Intermutualiste (IMA), Bruxelles, Belgium
- 10 3 Département Représentation et Etudes des Mutualités Libres, Bruxelles, Belgium
- 11 4 Centre de recherche en économie de la santé, gestion des institutions de soins et sciences
- 12 infirmières, Ecole de santé publique, Université libre de Bruxelles (U.L.B.), Bruxelles, Belgium
- 13 5 Service de Gynécologie-Obstétrique, Hôpital Universitaire de Bruxelles, Erasme, Université Libre de
- 14 Bruxelles (ULB), 808 route de Lennik, B-1070, Bruxelles, Belgium
- 15 6 Service études des Mutualités Socialistes, Bruxelles, Belgium
- 16 7 Pharmacologie Médicale, Faculté de Médecine, Université de Toulouse III, Inserm CERPOP, CHU,
- 17 Toulouse, France

18 Correspondence

- 19 Lionel Larcin, Centre de recherche Epidémiologie, Biostatistique et Recherche Clinique, Ecole de Santé
- 20 Publique, Université libre de Bruxelles (U.L.B.), Route de Lennik 808, CP 596, 1070 Bruxelles, Belgium
- 21 Email: lionel.larcin@ulb.be
- 22 Telephone: +32 2 555 40 71
- 23 Lionel Larcin: <u>https://orcid.org/0000-0002-5662-6788</u>

25 Abstract

Purpose: To investigate the trends and regional variations in uterotonics dispensed around birth
 between 2003 and 2018 in Belgium.

Methods: Data, including outpatient and inpatient prescriptions were extracted from a nationally representative prescription database. The prevalence of uterotonics dispensed during a period including the 7 days before birth, the delivery day and the 7 days after birth was computed over three 4-year-long study periods from 2003 to 2018. The trends between periods and associations between the use of at least one uterotonic and maternal age, region of residence, delivery type and social status were assessed using logistic regression.

Results: In total, 31,675 pregnancies were included in the study. The proportion of pregnancies exposed to at least one uterotonic decreased significantly from 92.9% (95%Cl, 92.3-93.4) in 2003–2006 to 91.4% (95%Cl, 90.7-92.0) in 2015-2018 for vaginal births and from 95.5% (95%Cl, 94.5-96.4) to 93.7% (95%Cl, 92.6-94.7) for caesarean sections. However, for vaginal births, the proportion of oxytocin increased from 84.5% (95%Cl, 83.7-85.2) to 89% (95%Cl 88.3-89.7). A significant association was found between uterotonic agent use and maternal age, region of residence, and delivery type. The dispensation of some uterotonic agents differed significantly between the regions.

41 Conclusions: The proportion of pregnancies exposed to at least one uterotonic was high across the
42 study period but decreased slightly between 2003 and 2018. Important variations in uterotonic use
43 between regions highlight the need for improved national guidance.

Keywords: administrative healthcare database, pregnancy, uterotonic, labour, induction,
augmentation, post-partum haemorrhage

46 Plain Language Summary:

This study investigated the trends and regional variations in uterotonics dispensed around birth between 2003 and 2018 in Belgium. Most pregnancies in Belgium involve exposure to at least one uterotonic around birth. Between 2003 and 2018, the proportion of pregnancies exposed to at least one uterotonic slightly decreased, suggesting a trend favoring a less medicalized birth experience in 51 Belgium. However, the use of some uterotonics has increased, as has oxytocin and misoprostol use in 52 vaginal birth. We also observed significant differences in the proportion of pregnancies exposed to 53 different uterotonics across the three regions of Belgium, emphasizing the need for uniform national 54 guidelines.

55 Key Points:

The proportion of pregnancies exposed to at least one uterotonic around birth decreased
 slightly between 2003 and 2018.

The use of at least one uterotonic agent was associated with maternal age, region of
 residence, and delivery type.

• The off-label use of misoprostol in obstetrics increased significantly between 2003 and 2018.

In the period (2015-2018), 89% of pregnancies were exposed to at least one oxytocin
 prescription for a vaginal birth.

• Important variations were observed between regions of uterotonic dispensation.

64

65 Introduction

In obstetrics, uterotonics are used in various situations, including abortion, labor induction, 66 and the prevention and treatment of postpartum hemorrhage (PPH). Despite their benefits, 67 uterotonics may also be associated with adverse events^{1,2} such as tachycardia, high blood pressure, 68 vomiting, and possibly impact breastfeeding rates.^{3,4}A study conducted in Sweden has observed that 69 70 among 1,267 pregnant women, the induction or augmentation of labor with oxytocin was used in 55% 71 of pregnancies while the frequency of labor dystocia was 19.8%. Additionally, incorrect dosages of 72 oxytocin were administered, with 7.3% of pregnant women receiving a higher dose and 42.6% a lower dose than the one recommended by guidance.⁵ Therefore, monitoring their use is important and can 73 74 play a role in correcting uterotonic prescription habits in obstetrics.

The risk-benefit balance is different when uterotonic agents are used before delivery to induce
 and augment uterine contractions than when used after birth to prevent and treat PPH associated with

77 uterine atony. Induction and augmentation of labor often involve the use of uterotonics. The World Health Organization (WHO) recommends the induction of labor for women who are known to have 78 reached 41 weeks of gestation.⁶ Although uterotonics have long been known to effectively induce or 79 augment labor,⁷⁻⁹ their safety still cannot be guaranteed. Some studies have reported an association 80 between the use of uterotonics and a higher risk of PPH¹⁰⁻¹² and an increased risk of stillbirth and 81 neonatal asphyxia.¹³ They should only be used when continuing pregnancy involves more risks than 82 uterotonic use.⁶ Uterotonics are also used in the third stage of labor to prevent or treat PPH after 83 84 vaginal birth (VB) or caesarean section (CS). PPH is an important cause of maternal death.¹⁴ Most 85 guidelines promote the administration of uterotonics immediately after birth¹⁵⁻¹⁹ as part of "the active 86 management" of the third stage of labor, including cord clamping and controlled cord traction to help deliver the placenta. Although active management reduces maternal blood loss at birth,²⁰ it may also 87 88 increase maternal diastolic blood pressure, postpartum vomiting, postpartum pain, and hospital readmission due to bleeding.²⁰ 89

In other countries, studies using large databases or billing data to evaluate the use of
 uterotonics are scarce because such databases often do not capture medications used in hospitals.²¹⁻
 ²³ In Belgium, inpatient medications are recorded; therefore, Belgian data represent an opportunity to
 highlight practices in real-world contextual settings. The present study explored at national and
 regional level the trends and prescribing pattern of uterotonics dispensed around birth in Belgium.
 More specifically, we assessed the period of exposure in the seven days preceding childbirth, the day
 of childbirth and seven days after childbirth.

97

98 Methods:

99 Study design and data source

100 This was a retrospective drug utilization study. Data were extracted from the permanent 101 sample (EPS). In Belgium, health insurance is mandatory, 98% of residents are captured. The EPS 102 database is a 2.5% representative sample constituted by the Inter Mutualistic Agency with the 103 information received by all insurance funds. The information is collected by patient and includes a 104 pseudonymised unique patient identifier, demographic characteristics such as patient's age and 105 residence region. The social status can also be identified because patients with low-income benefit 106 from a preferential reimbursement rate. Medications prescribed and dispensed from community and 107 hospital pharmacies were captured. In community pharmacies, only reimbursed medications were 108 registered. For hospital pharmacies, all medications including non-reimbursed medications prescribed 109 and dispensed during hospitalization were captured even for a day-care stay at the hospital or in 110 ambulatory care. Additionally, the medication received when the patient left the hospital was also 111 recorded. Information collected on medication includes classification according to the Anatomical Therapeutic Chemical Classification Code (ATC), the exact date of dispensation, and the quantity 112 113 dispensed. All information was completely anonymized and accessible for research purposes under strict conditions. ²⁴ 114

115 Study periods

To examine the evolution in the prescriptions of uterotonics dispensed around birth, we established three study periods of four years each between 2003 and 2018: (2003–2006) (2009–2012) (2015–2018).

119 Study population

This analysis only considered women who gave birth. Reimbursement delivery-only codes were used to select women from the EPS. The selection of codes only included deliveries that occurred after the 180th day of pregnancy.²⁵ Mothers whose data were not available in the EPS for the entire pregnancy period and those whose residences were outside Belgium were excluded. Finally, we excluded self-employed mothers who did not benefit from the same reimbursement scheme for drugs during the first study period (2003–2006) for all three periods.

126 Definition of exposure and measurement

127 We identified all ATC codes associated with a uterotonic used to induce labor or to prevent or 128 treat PPH commercialized in Belgium during the study period: dinoprostone (ATC: G02AD02), 129 carboprost (ATC: G02AD04), methylergometrine (ATC: G02AB01), oxytocin (ATC: H01BB02), carbetocin (ATC: H01BB03), and misoprostol (ATC: A02BB01-G02AD06). The list is presented in Supplementary 130 131 Table S1. We did not include mifepristone (ATC: G03XB01) because this medication is mainly used to 132 manage miscarriages and fetal death. We considered the period of exposure to seven days preceding childbirth, the day of childbirth, and seven days after childbirth, as uterotonics might be prescribed 133 134 and dispensed before, during, and after delivery.

135 Statistical methods

For the three study periods, the proportion and 95% confidence intervals of pregnancies exposed to at least one uterotonic from the pre-established list and for each individual uterotonic were computed. We also computed the proportion of pregnancies exposed to at least two and three distinct subgroups of uterotonics (different ATC codes at the 5th level). The results are presented separately for the VB and CS. Logistic regression analysis was used to assess the trends in the proportion of pregnancies exposed to uterotonics across the three study periods, adjusted for maternal age.

For the last study period, we computed the adjusted odds ratio using logistic regression to measure the strength of the association between the proportion of pregnancies exposed to at least one uterotonic agent and maternal age, region of residence, delivery type, and social status.

Additionally, to explore regional disparities in the proportion of pregnancies exposed to different uterotonics, we assessed the proportions by region and presented the results separately for VB and CS. We determined significant differences between regions after adjusting for maternal age using logistic regression analysis.

149

151 Results

152 This study included 31,675 pregnancies during the three study periods (Figure 1). Table 1 153 shows the proportion of maternal age, region of residence of the mother, and delivery type.

154 In the period 2003–2006, the majority of mothers were in the age group 25–29 years while in 155 the period 2015–2018 the majority were in the age group 30–34 years.

156 In the last study period, 19.1% of pregnant women benefited from a preferential 157 reimbursement rate.

158 The prevalence of pregnancies exposed to at least one, two, or three subgroups of uterotonics 159 (different ATC codes at the 5th level) dispensed around birth is shown in Table 2. Between 2003 and 160 2018, decreases in the use of at least one and two or more different uterotonics (from 92.9% to 91.4% 161 and from 45.3% to 25.7%) were observed for VB. Similar significant decreases were observed for CS (from 95.5% to 93.7% and 38.8% to 29.9%) as well. The factors associated with pregnancies with 162 163 exposure to at least one uterotonic agent are listed in Table 3. Maternal age, region of residence, and 164 delivery type were statistically associated after adjustment. No association was found between social 165 status and the use of at least one uterotonic.

166 The proportion of pregnancies exposed to at least one prescription for each uterotonic is listed 167 in Table 4. For VB, the proportion of pregnancies exposed to oxytocin was high and increased 168 significantly across the three study periods. In contrast, oxytocin use for CS decreased during the three 169 study periods.

The proportion of pregnancies exposed to at least one prescription of different uterotonics during the three distinct periods around birth is listed in Table 5. The exposure period was divided into three distinct periods:7 days before delivery, the day of delivery, and 7 days after delivery. Most exposures occurred on the day of the delivery. Except for dinoprostone, pregnancies were exposed more frequently to each uterotonics seven days after delivery than seven days before delivery. The geographical variations in uterotonics dispensed around delivery are listed in Table 6. For VB, the results reflected a higher prevalence of pregnancies exposed to oxytocin in Wallonia. The proportion of misoprostol use was slightly higher in Flanders than in Brussels but was much lower in Wallonia. For CS, carbetocin use was similar in Flanders and Wallonia but was significantly lower in the Brussels region.

180 Discussion

The proportion of pregnancies exposed to at least one uterotonic decreased slightly across the three study periods. For the proportion of pregnancies exposed to two, three, or more subgroups of uterotonics, the decrease was more significant, which might be explained by the increasing trend in the use of uterotonics that fit several indications, such as misoprostol and oxytocin. In addition, we hypothesized that an important part of the proportion of pregnancies with two or more different uterotonics would reflect induced deliveries. The decrease observed for this proportion might be related to the decrease in induced labor observed in Belgium from 32.1% in 2002 to 26.6% in 2015.²⁶

188 Widely prescribed Oxytocin

189 In the last study period, 89% of VB were exposed to at least one prescription of oxytocin 190 dispensed around birth. The widespread use of oxytocin for VB is based on WHO recommendations 191 for PPH prevention. Although a very large consensus supports uterotonic use to prevent PPH^{15,16,27}, we 192 observed 11% of VB not exposed to oxytocin and 8.6% of pregnancies not exposed to any uterotonics. 193 Unexposed pregnancies may reflect, in part, the place of the "expectant management" in the third 194 stage of labor in Belgium. Indeed, in addition to active management involving the systematic use of 195 uterotonics, "expectant management" allows the placenta to be delivered spontaneously through maternal effort and gravity.²⁸ 196

197 The significant decrease observed in oxytocin use for CS was most likely related to the 198 introduction of carbetocin (Pabal[®]) in Belgium in 2008. Our results suggest that carbetocin could replace oxytocin to prevent PPH after CS. Similar results were observed in a study conducted in
 Canada.²¹

In the last study period, we examined prescriptions dispensed during three different exposure periods: (a) seven days before delivery, (b) day of delivery, and (c) seven days after delivery. This separation aimed to distinguish the uterotonics used to induce labor from those used to prevent or treat PPH. Oxytocin was the most represented uterotonic in the period before delivery and on the day of delivery, suggesting that it was the most commonly used uterotonic to induce or augment labor.

206 A drastic decrease in methylergometrine use

207 In Belgium, methylergometrine was commercialized only under the name Methergin[®], and 208 was largely used in the first study period (2003–2006). However, a drastic decrease was observed 209 during the next two study periods. This is a consequence of several cases of accidental oral 210 administration of Methergin® to the newborn because of confusion between the Methergin® 211 dedicated to the mother and the paediatric preparation (often vitamin K) dedicated to the newborn.^{29,30} Because of these incidents, drop preparations of Methergin[®] were withdrawn from the 212 Belgian market in 2011.³¹ Only intramuscular administration of Methergin[®] remained available. Our 213 214 study suggests that for PPH prevention in VB, Methergin[®] has been replaced by oxytocin. For CS, our 215 data suggest replacement of Methergin[®] with carbetocin.

216 The off-label Misoprostol use: an increasing trend

We observed an increasing trend in the misoprostol prescribed and dispensed around birth between 2003 and 2018. In Belgium, until 2016, misoprostol was only available under the Cytotec[®] formulation, which is not approved for use in obstetrics.³² The increasing trend of Cytotec[®] use observed in our study may be explained by several factors. Many studies over the past two decades have suggested that misoprostol effectively induces labor and treats PPH.^{8,33-35} Additionally, misoprostol is easy to store at room temperature, inexpensive, and has a short half-life.^{35,36} The offlabel use of misoprostol is controversial. In 2005 and 2013, the French National Agency for Medicine issued warnings about the risks associated with Cytotec[®] use in obstetrics and gynaecology³⁷ and the
drug was withdrawn from the French market in 2018.³⁸ More recently, in March 2020, the German
Federal Institute for Drugs and Medical Devices was informed of new reports of severe side effects
when using Cytotec[®] outside the approved indication.³⁹

228 Geographical variation

Belgium is divided into three regions: Flanders, Wallonia, and Brussels. While the accessibility and reimbursement status of different uterotonics were the same across the country, we observed wide variations in their prescription patterns. Adjusted for maternal age, delivery type and social status our results found a statistically significant association between region of residence and higher odds of uterotonic dispensation. These variations may be explained by various factors.

First, important variations in obstetric practices between the regions of Belgium have influenced the use of uterotonics. For example, the proportion of induced deliveries varies significantly depending on region. In 2017, the proportion of induced deliveries was 31.6% in Wallonia⁴⁰, 24.6% in Flanders⁴¹, and 28.6% in the Brussels region.⁴²

The reimbursement status may also influence the choice of uterotonics. For example, carbetocin (Pabal®) was used much less frequently in the Brussels region than in the other regions in our study. This drug is not reimbursed and is relatively expensive.⁴³ Some hospitals might prefer to use oxytocin which is reimbursed for preventing uterine atony, instead of Pabal®. The cost may also have an important influence. We observed that the use of misoprostol (Cytotec®) was approximately twice as high in the Brussels region and Flanders compared to that in Wallonia, which might be explained by the low cost of misoprostol (Cytotec®).^{44,45}

A study conducted in Sweden reported significant differences in the rate of oxytocin use during labor in different hospitals.¹ According to the authors, these variations might be due to differences in "delivery culture" between hospitals. Changes in patient demand and expectations may explain some of these variations. We observed a significantly lower rate of oxytocin use in the Brussels region. One hypothesis to explain this result is lower adherence to the recommended routine uterotonics use to prevent PPH in the Brussels region than in Flanders and Wallonia. Several initiatives for low-risk births have recently been established to respond to this demand for less medicalized births in Brussels.⁴⁶⁻⁴⁸ For example, a new birth center has been established in a large hospital in Brussels to provide a package of care for uncomplicated pregnancies, where the minimisation of medical interventions is encouraged.⁴⁸

255 Differences in pharmaceutical marketing strategies between regions may explain the observed 256 variations. For example, the increased use of carbetocin, a relatively new uterotonic agent in the 257 Belgian market, might result from pharmaceutical marketing efforts in hospitals that are more 258 sensitive to innovation.

Educational factors might also explain the variation; healthcare workers' education is not equivalent between the different regions in Belgium. Different regional ministries of education and training are responsible for determining the policies of the education system.

Finally, the variations observed reflect the lack of clear national evidence-based guidelines for the use of uterotonics around birth in Belgium. Consequently, the French-speaking portion of Belgium might refer to guidance from France or Switzerland. In contrast, the Flemish-speaking portion might use guidelines from Anglo-Saxon countries or the Netherlands, impacting clinical practices.

266 Strengths and Limitations

For the first time, this study reports the patterns of uterotonic prescriptions in Belgium. Our sample was representative of the people enrolled in the health security system (98% of the population). We excluded self-employed pregnant women from all three periods because they did not benefit from the same reimbursement scheme throughout the study period. However, we compared the proportions with and without them in the last study period and obtained similar results for uterotonic use. Therefore, we do not believe that it has affected the representativeness of our sample. 273 Multiparity might influence the use of uterotonics, but because the data were not available 274 before 2003, we were not able to quantify multiparity completely. For the last study period (2015-275 2018), we checked the number and the proportion of pregnancies with at least one previous pregnancy 276 from the same mother after the year 2003 considering the years between the study periods. We 277 identified 5,163 (50.1%) multiparous pregnancies in the period. We compared the exposure of at least 278 one uterotonic among multiparous pregnancies to the other pregnancies and we found a lower rate 279 of uterotonic use among multiparous pregnancies (90.6 vs 93.1%). Therefore, multiparity may have 280 slightly impacted our results, this factor should be explored in future studies.

The information regarding the exact indications for uterotonics was missing. Our analyses assessed three periods of exposure to distinguish between uterotonics dispensed for labor induction and those dispensed to prevent PPH. However, in the proportion of pregnancies exposed on the day of delivery, we captured prescriptions of uterotonics dispensed for all indications: induction of labor and prevention of PPH. Therefore, any interpretation of indications and conclusions regarding misuse and overuse should be considered with extreme caution.

Finally, there were no validation studies inherent to the measurement of uterotonics with Belgian claims data; however, other studies have shown that claims databases were highly accurate in tracking uterotonics coverage compared to the survey report.⁴⁹ We have also presented our results to a team with expertise in gynecology to detect potential clinical contradictions of our results. Additionally, because our data were collected for the primary purpose of billing and reimbursement, controls were performed at different stages to detect inconsistencies in the data, which contributed to their quality.⁵⁰

294 Conclusion

295 Most pregnancies in Belgium involve exposure to at least one uterotonic around birth. 296 Between 2003 and 2018, the proportion of pregnancies exposed to at least one uterotonic decreased 297 slightly, suggesting a trend for less medicalized birth experience in Belgium. However, the proportion of some uterotonics has increased, as with oxytocin and misoprostol use in VB. We also observed significant differences in the proportion of pregnancies exposed to different uterotonics across the three regions of Belgium, emphasizing the need for uniform national guidelines.

301 Acknowledgements:

We would like to thank the InterMutalist Agency (IMA) for supplying the data and helping us with data management. Additionally, we would like to thank Editage Editing Services and Marianne Cuellar for

304 English language editing.

305 Author Contributions:

LL analyzed the data and drafted the manuscript. XR and GK contributed to data acquisition and interpretation. CDM, PVW, CL, and BD contributed to the interpretation of data and revisions of the manuscript. FKS formulated the objectives and design of the study, supervised the statistical analysis, interpreted the results, and revised the manuscript. All authors were involved in revising the manuscript and approved the final version of the manuscript.

311 Ethics Statement:

312 The authors state that no ethical approval was needed.

313 Patient consent statement:

- 314 Data were deidentified so individual informed consent was not needed.
- 315 Funding:

The authors received no financial support for the research, authorship, and/or publication of this

317 article.

318 Data availability statement:

319 Data that support the findings of this study are available from the InterMutalist Agency (IMA).

- 320 Restrictions apply to the availability of these data, which were used under license for this study. Data
- 321 are available at https://metadata.ima-aim.be/fr/app/bdds/Ps with permission from the InterMutalist
- 322 Agency (IMA).

323 References

324 1. Oscarsson ME, Amer-Wåhlin I, Rydhstroem H, Källén K. Outcome in obstetric care related to 325 oxytocin use. A population-based study. Acta Obstet Gynecol Scand. 2006;85(9):1094-1098. 326 2. Berglund S, Grunewald C, Pettersson H, Cnattingius S. Severe asphyxia due to delivery-327 related malpractice in Sweden 1990-2005. Bjog. 2008;115(3):316-323. 328 3. Jordan S, Emery S, Watkins A, Evans JD, Storey M, Morgan G. Associations of drugs routinely 329 given in labour with breastfeeding at 48 hours: analysis of the Cardiff Births Survey. Bjog. 330 2009;116(12):1622-1629; discussion 1630-1622. 331 4. McDonald S, Abbott JM, Higgins SP. Prophylactic ergometrine-oxytocin versus oxytocin for 332 the third stage of labour. Cochrane Database Syst Rev. 2004;2004(1):Cd000201. 333 Selin L, Almström E, Wallin G, Berg M. Use and abuse of oxytocin for augmentation of labor. 5. 334 Acta Obstet Gynecol Scand. 2009;88(12):1352-1357. 335 6. Organization WH. WHO recommendations: induction of labour at or beyond term. World 336 Health Organization; 2018. Frigoletto FD, Jr., Lieberman E, Lang JM, et al. A clinical trial of active management of labor. N 7. 337 338 Engl J Med. 1995;333(12):745-750. 339 8. Alfirevic Z, Aflaifel N, Weeks A. Oral misoprostol for induction of labour. Cochrane Database 340 Syst Rev. 2014;2014(6):Cd001338. 341 9. Thomas J, Fairclough A, Kavanagh J, Kelly AJ. Vaginal prostaglandin (PGE2 and PGF2a) for 342 induction of labour at term. Cochrane Database Syst Rev. 2014;2014(6):Cd003101. 343 10. Belghiti J, Kayem G, Dupont C, Rudigoz RC, Bouvier-Colle MH, Deneux-Tharaux C. Oxytocin 344 during labour and risk of severe postpartum haemorrhage: a population-based, cohort-345 nested case-control study. BMJ Open. 2011;1(2):e000514. 346 11. Davey MA, Flood M, Pollock W, Cullinane F, McDonald S. Risk factors for severe postpartum 347 haemorrhage: A population-based retrospective cohort study. Aust N Z J Obstet Gynaecol. 348 2020;60(4):522-532. 349 12. Erickson EN, Carlson NS. Predicting Postpartum Hemorrhage After Low-Risk Vaginal Birth by 350 Labor Characteristics and Oxytocin Administration. J Obstet Gynecol Neonatal Nurs. 351 2020;49(6):549-563. 352 13. Mohan SB, Sommerfelt H, Frøen JF, et al. Antenatal Uterotonics as a Risk Factor for 353 Intrapartum Stillbirth and First-day Death in Haryana, India: A Nested Case-control Study. 354 Epidemiology (Cambridge, Mass). 2020;31(5):668. 355 14. Khan KS, Wojdyla D, Say L, Gülmezoglu AM, Van Look PF. WHO analysis of causes of maternal 356 death: a systematic review. Lancet. 2006;367(9516):1066-1074. 357 15. Dahlke JD, Mendez-Figueroa H, Maggio L, et al. Prevention and management of postpartum 358 hemorrhage: a comparison of 4 national guidelines. Am J Obstet Gynecol. 359 2015;213(1):76.e71-76.e10. 360 16. Organization WH. WHO recommendations Uterotonics for the prevention of postpartum haemorrhage: Web annex 7: Choice of uterotonic agents. World Health Organization;2018. 361 17. 362 Organization WH. WHO recommendations on maternal health: guidelines approved by the 363 WHO Guidelines Review Committee. World Health Organization;2017. Prendiville W, Elbourne D, Chalmers I. The effects of routine oxytocic administration in the 364 18. 365 management of the third stage of labour: an overview of the evidence from controlled trials. Br J Obstet Gynaecol. 1988;95(1):3-16. 366 19. 367 Mannaerts D, Van der Veeken L, Coppejans H, Jacquemyn Y. Adverse Effects of Carbetocin 368 versus Oxytocin in the Prevention of Postpartum Haemorrhage after Caesarean Section: A 369 Randomized Controlled Trial. J Pregnancy. 2018;2018:1374150. 370 20. Begley CM, Gyte GM, Devane D, McGuire W, Weeks A, Biesty LM. Active versus expectant 371 management for women in the third stage of labour. Cochrane Database Syst Rev. 372 2019;2(2):Cd007412. Thorneloe B, Carvalho JCA, Downey K, Balki M. Uterotonic drug usage in Canada: a snapshot 373 21. 374 of the practice in obstetric units of university-affiliated hospitals. Int J Obstet Anesth. 375 2019;37:45-51.

376	22.	Sabaté M, Ferrer P, Ballarín E, et al. Inpatient drug utilization in Europe: nationwide data
377		sources and a review of publications on a selected group of medicines (PROTECT project).
378		Basic Clin Pharmacol Toxicol. 2015;116(3):201-211.
379	23.	Ferrer P, Ballarín E, Sabaté M, et al. Sources of European drug consumption data at a country
380		level. Int J Public Health. 2014;59(5):877-887.
381	24.	L'Agence InterMutualiste. Echantillon Permanent <u>https://metadata.ima-</u>
382		aim.be/fr/app/bdds/Ps. Accessed 20/08/2022.
383	25.	INAMI.Nomenclature article 9.
384	https:/	/www.inami.fgov.be/fr/nomenclature/nomenclature/Pages/nomen-article09.aspx, 2022.
385	26.	STATBEL. Naissances et fécondité https://statbel.fgov.be/fr/themes/population/naissances-
386		et-fecondite. Accessed 20/08/2022.
387	27.	Evensen A, Anderson JM, Fontaine P. Postpartum Hemorrhage: Prevention and Treatment.
388		Am Fam Physician. 2017;95(7):442-449.
389	28.	Begley CM, Guilliland K, Dixon L, Reilly M, Keegan C. Irish and New Zealand midwives'
390		expertise in expectant management of the third stage of labour: the 'MEET' study. <i>Midwifery</i> .
391		2012;28(6):733-739.
392	29.	Davanzo F, Settimi L, Giordano F, Casini ML, Ferrazin F. Inadvertent oral administration of
393		methylergometrine maleate to children in the first months of life: from surveillance to
394		prevention. Pharmacoepidemiol Drug Saf. 2015;24(3):269-275.
395	30.	Bangh SA, Hughes KA, Roberts DJ, Kovarik SM. Neonatal ergot poisoning: a persistent
396		iatrogenic illness. Am J Perinatol. 2005;22(5):239-243.
397	31.	Novartis. Direct Healthcare Professional Communications.
398		https://www.afmps.be/sites/default/files/downloads/Methergin%202011%2010%2024%20F
399		R.pdf.
400	32.	Song J. Use of misoprostol in obstetrics and gynecology. Obstet Gynecol Surv.
401		2000;55(8):503-510.
402	33.	Weeks AD, Fiala C, Safar P. Misoprostol and the debate over off-label drug use. Bjog.
403		2005;112(3):269-272.
404	34.	Amini M, Reis M, Wide-Swensson D. A Relative Bioavailability Study of Two Misoprostol
405		Formulations Following a Single Oral or Sublingual Administration. Front Pharmacol.
406		2020;11:50.
407	35.	Alfirevic Z, Weeks A. Oral misoprostol for induction of labour. <i>Cochrane Database Syst Rev.</i>
408		2006(2):Cd001338.
409	36.	Fiala C, Gemzell-Danielsson K, Tang OS, von Hertzen H. Cervical priming with misoprostol
410		prior to transcervical procedures. Int J Gynaecol Obstet. 2007;99 Suppl 2:S168-171.
411	37.	ansm. Mise en garde sur les risques potentiels liés à l'utilisation hors AMM du Cytotec
412		(misoprostol) dans le déclenchement de l'accouchement et toute autre utilisation
413		gynécologique - Point d'information.
414	38.	ansm. Cytotec (misoprostol) : arrêt de commercialisation à compter du 1er mars 2018 -
415		Communiqué. https://archiveansm.integra.fr/S-informer/Communiques-Communiques-
416		Points-presse/Cytotec-misoprostol-arret-de-commercialisation-a-compter-du-1er-mars-
417		2018-Communique.
418	39.	BfArM. Cytotec (Misoprostol): Risiken im Zusammenhang mit einer Anwendung zur
419		Geburtseinleitung außerhalb der Zulassung ("off-label-use").
420		https://www.bfarm.de/SharedDocs/Risikoinformationen/Pharmakovigilanz/EN/RHB/2020/rh
421		<u>b-cytotec.pdf?</u> blob=publicationFile&v=2.
422	40.	Leroy Ch DC, Debauche Ch, Van Leeuw V Santé périnatale en Wallonie – Année 2017.
423		Centre d'Épidémiologie Périnatale;2019.
424	41.	Devlieger R ME, Goemaes R, Cammu H. Perinatale activiteiten in Vlaanderen 2017. Brussel:
425		Studiecentrum voor perinatale Epidemiologie,;2018.

426	42.	Van Leeuw V DC, Debauche Ch, Leroy Ch. Santé périnatale en Région bruxelloise –								
427		Années 2017. Centre d'Épidémiologie Périnatale;2018.								
428	43.	REPERTOIRE COMMENTE DES MEDICAMENTS 2018. 2018.								
429		https://www.cbip.be/ggr_pdfs/GGR_FR_2018.pdf.								
430	44.	Voigt F, Goecke TW, Najjari L, Pecks U, Maass N, Rath W. Off-label use of misoprostol for								
431		labor induction in Germany: a national survey. Eur J Obstet Gynecol Reprod Biol.								
432		2015;187:85-89.								
433	45.	Krause E, Malorgio S, Kuhn A, Schmid C, Baumann M, Surbek D. Off-label use of misoprostol								
434		for labor induction: a nation-wide survey in Switzerland. <i>Eur J Obstet Gynecol Reprod Biol.</i>								
435		2011;159(2):324-328.								
436	46.	Sud HI. <u>https://www.his-izz.be/fr/hopitaux-iris-sud/services-medicaux/maternite</u> 82, 2022.								
437	47.	Physalis_CHU_BRUGMAN. Samenwerking : Physalis in het UVC Brugmann. https://www.chu-								
438		brugmann.be/nl/news/20170301-physalis-vbov.asp.								
439	48.	Welffens K, Derisbourg S, Costa E, et al. The "Cocoon," first alongside midwifery-led unit								
440		within a Belgian hospital: Comparison of the maternal and neonatal outcomes with the								
441		standard obstetric unit over 2 years. <i>Birth.</i> 2020;47(1):115-122.								
442	49.	Tahsina T, Hossain AT, Ruysen H, et al. Immediate newborn care and breastfeeding: EN-								
443		BIRTH multi-country validation study. BMC pregnancy and childbirth. 2021;21(1):1-17.								
444	50.	inami. Offices de tarification : Instructions pour facturer les prestations pharmaceutiques								
445		https://www.inami.fgov.be/fr/professionnels/autres/offices-tarification/Pages/default.aspx.								
446										

447 Tables:

	Period 2003–2006	Period 2009–2012	Period 2015–2018
	(N = 10,357)	(N = 11,019)	(N = 10,299)
Variables	% (n)	% (n)	% (n)
Maternal age			
< 25 years	16 (1,660)	14.9 (1,636)	11.3 (1,165)
25-29 years	35.5 (3,678)	34.3 (3,779)	32.8 (3,381)
30-34 years	32.7 (3,389)	33.2 (3,660)	35.5 (3 <i>,</i> 660)
35-39 years	13 (1,343)	14.2 (1,564)	16.2 (1,670)
≥ 40 years	2.8 (287)	3.4 (380)	4.1 (423)
Region of residence			
Flanders	46.6 (4,823)	47.9 (5,277)	49 (5,051)
Wallonia	40.8 (5,225)	39.2 (4,323)	37.8 (3,898)
Brussels region	12.6 (1,309)	12.9 (1,416)	13.1 (1,348)
Delivery type			
Any vaginal birth	81 (8,394)	80.2 (8,833)	78.8 (8,117)
Cesarean section	19 (1,963)	19.8 (2,186)	21.2 (2,182)

Table 2. Prevalence of pregnancies exposed to at least one, two and three different uterotonics agent of the preestablished list dispensed around delivery in Belgium

	Va	ginal birth	Cesare	an section		
	n/N	% (95%CI)	n/N	% (95%CI)		
Proportion of pregnancies	exposed to at lea					
Period 2003-2006	7,798/8,394	92.9 (92.3-93.4)	1,875/1,963	95.5 (94.5-96.4)		
Period 2009-2012	8,198/8,833	92.8 (92.2-93.3)	2,066/2,186	94.5 (93.5-95.4)		
Period 2015-2018	7,419/8,117	91.4 (90.7-92)	2,045/2,182	93.7 (92.6-94.7)		
p-value for trends *		< 0.001		0.011		
Proportion of pregnancies exposed to at least two different** uterotonic agents						
Period 2003-2006	3,801/8,394	45.3 (44.2-46.3)	762/1,963	38.8 (36.6-41)		
Period 2009-2012	2,804/8,833	31.7(30.8-32.7)	677/2,186	30.1 (29-32.9)		
Period 2015-2018	2,084/8,117	25.7 (24.7-26.6)	639/2,182	29.9 (27.4-31.2)		
p-value for trends *		< 0.001		<0.001		
Proportion of pregnancies	exposed to at lea	st three different** uter	otonic agents			
Period 2003-2006	845/8,394	10.1 (9.4-10.7)	146/1,963	7.4 (6.3-8.7)		
Period 2009-2012	497/8,833	5.6 (5.1-6.1)	151/2,186	6.9 (5.9-8)		
Period 2015-2018	271/8,117	3.3 (2.9-3.7)	182/2,182	8.3 (7.2-9.6)		
p-value for trends *		<0.001		0.23		
* The second state state of the state of a state of the second	and a state of a different state	and the state of a set				

* Trend test using logistic regression adjusted on maternal age

** Considered different if ATC code at the 5th level is different

Period of exposure considered: the 7 days preceding childbirth, the day of childbirth and the seven days after childbirth all together.

455

456									
Table 3. Factors associated to pregnancies exposed to at least one uterotonic agent (2015-									
2018) N=10,297									
	n/N (%)	Adj OR 95 (%Cl)	p value						
Maternal age			0.01						
< 25 years	1,099/1,165 (94.3)	1.39 (1.06-1.1.85)							
25-29 years	3,174/3,381 (92.1)	1 (Ref)							
30-34 years	3,334/3,660 (91.1)	0.88 (0.74-1.04)							
35-39 years	1,524/1,670 (91.3)	0.90 (0.73-1.12)							
≥ 40 years	393/423 (92.9)	1.13 (0.76-1.68)							
Region of residence									
Flanders	4,619/5,051 (91.4)	1.63 (1.36-1.96)							
Wallonia	3,674/3,898 (94.2)	2.47 (2.01-3.05)	462						
Brussels region 1,169/1,348 (86.7) 1 (Ref)									
Mode of delivery <0.00									
Any vaginal birth	Any vaginal birth 7,419/8,117 (91.4) 1 (Ref)								
Cesarean section	2,045/2,182 (93.7)	1.42 (1.17-1.72)							
Social Status*			0.87						
No preferential rate	7,661/8,333 (91.9)	1 (Ref)	465						
Preferential rate 1,803/1,966 (91.7) 0.98 (0.74-1.04)									
Adj OR: Adjusted Odd Ratio from logistic regression model									
Period of exposure considered: the 7 days preceding childbirth, the day of childbirth and the									
seven days after childbirth altogether.									
*Social status: Pregnant women with a preferential rate is a proxy of low social status									

468

469

470

Table 5. Prevalence of pregnancies exposed to at least one prescription of uterotonic dispensed during three different periods of exposure around birth in 2015-2018 (N=10,299)

	7 days	before DD	Day	of delivery	7 days after DD				
	n	% (95%CI)	n	% (95%CI)	n	% (95%CI)			
oxytocin	987	9.6 (9-10.2)	6,472	62.8 (61.9-63.8)	1,508	14.6 (14-15.3)			
dinosprotone	585	5.7 (5.2-6.1)	1,119	10.9 (10.3-11.5)	345	3.3 (3-3.7)			
methylergometrine	13	0.1 (0.1-0.2)	194	1.9 (1.6-2.2)	52	0.5 (0.4-0.7)			
carboprost	16	0.2 (0.1-0.2)	87	0.8 (0.7-1)	23	0.2 (0.1-0.3)			
carbetocin	22	0.2 (0.1-0.3)	733	7.1 (6.6-7.6)	57	0.6 (0.4-0.7)			
misoprostol	55	0.5 (0.4-0.7)	317	3.1 (2.7-3.4)	107	1 (0.8-1.2)			
DD : Day of Delivery									

Table 4. Prevalence of uterotonics prescribed and dispensed around birth between 2003 and 2018

			-		-		
		2003-2006		2009-2012		2015-2018	
Vaginal only		N=8,394		N=8,833		N=8,117	p-value for
	n	% (95%CI)	n	% (95%CI)	n % (95%CI)		trends *
oxytocin	7,089	84.5 (83.7-85.2)	7,837	88.7(88-89.4)	7,226	89 (88.3-89.7)	<0.001
dinosprotone	1,873	22.3 (21.4-23.2)	1,739	19.7 (18.9-20.5)	1,686	20.8 (19.9-21.7)	0.016
methylergometrine	3,156	37.6 (36.6-38.6)	1,382	15.6 (14.9-16.4)	193	2.4 (2.1-2.7)	<0.001
carboprost	63	0.8 (0.6-0.9)	80	0.9 (0.7-0.11)	91	1.1 (0.9-0.14)	0.011
carbetocin	NA	NA	55	0.6 (0.5-0.8)	82	1 (0.8-1.2)	NA
misoprostol	95	1.1 (0.9-1.4)	359	4 (3.7-4.5)	422	5.2 (4.7-5.7)	<0.001
Casaraan daliwaru	2003-2006		2009-2012			2015-2018	
Cesarean delivery	N=1,963		N=2,186			N=2,182	p-value for
only	n	% (95%CI)	n	% (95%CI)	n	% (95%CI)	trends *
oxytocin	1,824	92.9 (91.7-94)	1,835	83.9 (82.3-85.5)	1,536	70.4 (68.4-72.3)	<0.001
dinosprotone	29	15.2 (13.7-16.9)	311	14.2 (12.8-15.8)	316	14.5 (13-16)	0.75
methylergometrine	540	27.5 (25.5-29.5)	279	12.8 (11.4-14.2)	65	3 (2.3-3.8)	< 0.001
carboprost	16	0.8 (0.5-1.3)	31	1.4 (0.9-2)	33	1.5 (1-2.1)	0.047
carbetocin	NA	NA	339	15.5 (14-17.1)	729	33.4 (31.4-35.4)	NA
misoprostol	18	0.9 (0.5-1.4)	46	2.1 (1.5-2.8)	56	2.6 (1.9-3.3)	<0.001

* Trend test using logistic regression analyses adjusted on maternal age at birth.

NA = Not Applicable: Carbetocin was not yet commercialized in Belgium in the period 2003-2006.

Period of exposure considered: the 7 days preceding childbirth, the day of childbirth and the seven days after childbirth all together.

Table 6. Geographical variations in the proportion of pregnancies exposed to at least one uterotonic agent at the end of pregnancy in 2015-2018

Vaginal only	Flanders N=3,987		Wallonia N=3,066		Bruss	p-value	
	n	% (95%CI)	n	% (95%CI)	n	% (95%CI)	*
Oxytocin	3,532	88.6 (87.6-89.5)	2,827	92.2 (91.2-93.1)	866	81.5 (79-83.8)	< 0.001
Dinoprostone	871	21.8 (20.6-23.2)	650	21.2 (19.8-22.7)	165	15.5 (13.4-17.8)	< 0.001
Misoprostol-Cytotec	252	6.3 (5.6-7.1)	115	3.8 (3.1-4.5)	54	5.1 (3.8-6.6)	< 0.001
Methylergometrine	94	2.4 (1.9-2.9)	74	2.4 (1.9-3)	25	2.4 (1.5-3.4)	0.97
Carboprost	33	0.8 (0.6-1.2)	45	1.5 (1.1-1.9)	13	1.2 (0.6-2.1)	0.045
Carbetocin	51	1.3 (0.9-1.7)	30	1 (0.7-1.4)	<5	<0.1 ()	0.021
Cosaraan deliyany anly	Flanders N=1,064		Wallonia N=832		Brussels region N=285		p-value
	n	% (95%CI)	n	% (95%CI)	n	% (95%CI)	*
Oxytocin	731	68.7 (65.8-71.5)	594	71.4 (68.2-74.4)	210	73.7 (68.2-78.7)	0.15
Dinoprostone	145	13.6 (11.6-15.8)	125	15 (12.7-17.6)	46	16.1 (12.1-20.1)	0.47
Misoprostol-Cytotec	31	2.9 (2-4.1)	17	2 (1.2-3.2)	8	2.8 (1.2-5.5)	0.5
Methylergometrine	47	4.4 (3.3-5.8)	12	1.4 (0.7-2.5)	6	2.1 (0.8-4.5)	< 0.001
Carboprost	9	0.8 (0.4-1.6)	20	2.4 (1.5-3.7)	<5	<1.5 ()	0.035
Carbetocin	353	33.2 (30.3-36.1)	322	38.7 (35.4-42.1)	54	18.9 (14.6-24)	< 0.001

* P value from logistic regression adjusted on maternal age

Period of exposure considered: the 7 days preceding childbirth, the day of childbirth and the seven days after childbirth all together.



494

- 495 Figure:
- 496

Figure 1. Flowchart of the study

